a0 Thursday 187372010

ests for RB in sector 56 to 6 kA and to 2 kA for all other
), sectors completed by 12:00 = Green light for 6 KA operation

o Tests for 2 A/s ramp rate of the main quadrupoles completed
by 13:00. Two thresholds had to be increased to 600 uV

e PIC in Sector 12 back to nominal =» Tested

e SW limits for ramp rate to 2 A/s for RB

e 15:00 Access required for Beam Dump kicker (faulty power
supply)

e 15:30 Recovered but DB problem not allowing rearming the
beam dump

e 16:30 Finally ready for ramp without beam

o 17:38 Trip of Sector 78 during dry ramp to 6 kA. Not
understood
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Dump during the ramp due to BLM (problem with optical
—._for BLM in point 1 =>» fixed).

e 20:45 Try a second ramp without beam

o 22:04 Ramp to 3.5 TeV without beam completed
e 00:20 Ready at injection

e Injection at 01:45

* Found huge chromaticity (negative) and important orbit
distortion -> regenerated SF and SD functions (stuff up)->
much better!

 Performed checks:
— Bl:
e QTrim:H=-25.0 &V=0.0-NowatH=5,V=5
e Qtrim: Honlyby+0.02. Tunes at 0.28 and 0.31
* Orbit corrected against “golden™
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Performed checks:
— B2
e Q' trim: Hby-20.0-V by +8 - Now at H= 10, V=9
o Small tune trims to get back to nominal
e Orbit corrected < vias e :

Edit Run Timing Configure
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* REA: Ihcop Use

(‘info [ FFT | PLL | DataSets | @ | FB/Trim | orbit |

Data Set Selector

LHC - B1 - fill #3532 - O'-#0: dp=-+.00010 - LHC.BOBEQ.UA47 . FFT2_BL - 2010-03-13 .

Max. Number of Datasets: 20 =
0 - LHC 2010-03-19_03:07:57 - no comment % -504
m 1 - LHC 2010-03-19_03:08:07 - Q'-#0: dp=-+.00010 E -60
M 2 - LHC 2010-03-19_03:08:16 - Q'-#1: dp=+.00000 g -70
M 3 - LHC 2010-03-19_03:08:33 - Q'-#2 dp=+.00010 T 8O
5 4 - LHC 2010-03-19_03:08:07 - Q'-Result: +43.1/+3.7 E _gp -
|5 - LHC 2010-03-19_0%:08:55 - Q'-#0: dp=-+.00010 2 _100-
6 - LHC 2010-03-19_03:09:05 - Q'-#1: dp=+.00000 _1104

7 - LHC 2010-03-19_03:09:21 - Q'-#2: dp=+.00010 _1201

9 § - LHC 2010-03-19_03:08:55 - Q'-Result: +40.6/+4.8 aEnd

. . .
001 10 - LHC 2010-03-18_03:12:35 - Q'-#0: dp=- +.00010 0 o Gl Gy G2 s B9 R 08 A

el frequency [frev
"1 11 - LHC 2010-03-19_03:12:44 - Q'-#1: dp—+.00000 fuenay [frev)

1 12 - LHC 2010-03-19_03:13:00 - Q'-#2 dp=+.00010 MI ACQ#IIE Mise.

LHC - B1 - fill #9532 - Q'-#0: dp=-+.00010 - LHC BOBEQ.UA47 FFT2 EL - 2010-03-19

98 9 - LiC 2010-03-19_03:09:51 - no comment

M 13 - LHC 2010-03-19_03:12:35 - Q'-Result +5.3/+4.9

| Open || Save H Delete H Refresh

| | |
®m wo;
2 = 2

L L L

| Compose MNew Dataset |

-90+
Spawn TuneViewer Display

vertical amplitude [dB]

C s: =iy

|n0 comment

-110+
-120

' -130+

Q o
aute-save —1404
- . T T T T T T T T
@ @ @ -l O 0 005 01 015 02 025 03 035 04 045

frequency [frev]
R3:14:22 - 2= Ulconf PLL UpdateSettings() - updale read settings Il




“ Frid ay 19/3/2010

AN

04:00 : Ramp started with bucket 1 and 1001 with tune
. feedback on, LHCPROBE, ~5e9

e At 05:23:3.5TeV 1100 hrs lifetime

 At5:25: RQTF.A81B2, RQTF.A45B2, RQTF.A34B2,
RQTF.A23B2 tripped - PM indicates that QPS triggered

= YASP DV LHCRING / 3.5TeV 2Aps / beam 1 ==
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Beam 1, uncorrected, at 3.5 TeV
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Beam 2, uncorrected, at 3.5 TeV

Had time to send 50% of an orbit correction on B1 and B2, before
RQTFs tripped.




-0.5

orbit corrector trim [urad]

-1

tune trim [frev]

chroma trim [frev]

-1

EI- QP FixDisplay RT-Trims Evolution |

0.5

-0.01+
-0.02+
-0.03
-0.04
-0.05+
-0.06

Friday 19/3/2010

Save

Legend
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Tune Feedback unable to keep Beam 2

Tunes at nominal settings after the end of the ramp. Trims requested to RQT circuits
therefore increase & probably led to trip of QPS.

Probable explanation:

Coupling in Beam 2 reaches tune split, so pushes tunes apart. Tune feedback tries to
compensate but is unable to do so & requests stronger & stronger trims from the RQTs
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B1 emittance evolution during the first ramp to 3.5TeV

BSRT vs WS Vertical excellent up to ~ 2.5 TeV, then BSRT calibration

with D3 light to be studied

WS profiles with small sigmas to be looked in detalil

BSRT vs WS HOR withalready seen systematic difference, to be understood
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VLC media player
Fle View Settings Audio Video MNavigation Help
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LBDS <BEAM DUMPED >
ARMING

KICKER BETS IPOC

LASS RETRIG TSU
INJECTION
REMOTE ACOND. TIMING

STANDBY 0 min

BETVDD

LBDS <BEAM DUMPED >
ARMING
KICKER BETS IPOC
LASS RETRIG TSU
INJECTION
REMOTE — ACOND. TIMING

STANDBY 0 min

BETVDD

LHCBEAM1-19/03/2010 05:41:29.549 LHCBEAMZ2-19/03/2010 05:41:30.116

0:00:00f 0:00: |x1.00 |"LHC Beam Dump System"

Beams dumped properly
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~* Plans of the day:

— Morning:
* |Investigate / cure RQTF trips

o Operation at 450 GeV (aperture studies- damper commissioning
TBC)

— Afternoon: Next 3.5 TeV ramp trial
— Night: Optics measurements

e Over the week end: 3.5 TeV operation
— MPS and LBDS tests

— Protection devices and collimators setting-up
— Optics measurements
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